

A Service of

ZBW

Leibniz-Informationszentrum Wirtschaft Leibniz Information Centre for Economics

Tracey, Marlon R.; Polachek, Solomon

Working Paper Heterogeneous Layoff Effects of the US Short-Time Compensation Program

IZA Discussion Papers, No. 11746

Provided in Cooperation with: IZA – Institute of Labor Economics

Suggested Citation: Tracey, Marlon R.; Polachek, Solomon (2018) : Heterogeneous Layoff Effects of the US Short-Time Compensation Program, IZA Discussion Papers, No. 11746, Institute of Labor Economics (IZA), Bonn

This Version is available at: https://hdl.handle.net/10419/185206

Standard-Nutzungsbedingungen:

Die Dokumente auf EconStor dürfen zu eigenen wissenschaftlichen Zwecken und zum Privatgebrauch gespeichert und kopiert werden.

Sie dürfen die Dokumente nicht für öffentliche oder kommerzielle Zwecke vervielfältigen, öffentlich ausstellen, öffentlich zugänglich machen, vertreiben oder anderweitig nutzen.

Sofern die Verfasser die Dokumente unter Open-Content-Lizenzen (insbesondere CC-Lizenzen) zur Verfügung gestellt haben sollten, gelten abweichend von diesen Nutzungsbedingungen die in der dort genannten Lizenz gewährten Nutzungsrechte.

Terms of use:

Documents in EconStor may be saved and copied for your personal and scholarly purposes.

You are not to copy documents for public or commercial purposes, to exhibit the documents publicly, to make them publicly available on the internet, or to distribute or otherwise use the documents in public.

If the documents have been made available under an Open Content Licence (especially Creative Commons Licences), you may exercise further usage rights as specified in the indicated licence.



WWW.ECONSTOR.EU



Initiated by Deutsche Post Foundation

DISCUSSION PAPER SERIES

IZA DP No. 11746

Heterogeneous Layoff Effects of the US Short-Time Compensation Program

Marlon R. Tracey Solomon W. Polachek

AUGUST 2018



Initiated by Deutsche Post Foundation

DISCUSSION PAPER SERIES

IZA DP No. 11746

Heterogeneous Layoff Effects of the US Short-Time Compensation Program

Marlon R. Tracey Southern Illinois University Edwardsville

Solomon W. Polachek State University of New York at Binghamton and IZA

AUGUST 2018

Any opinions expressed in this paper are those of the author(s) and not those of IZA. Research published in this series may include views on policy, but IZA takes no institutional policy positions. The IZA research network is committed to the IZA Guiding Principles of Research Integrity.

The IZA Institute of Labor Economics is an independent economic research institute that conducts research in labor economics and offers evidence-based policy advice on labor market issues. Supported by the Deutsche Post Foundation, IZA runs the world's largest network of economists, whose research aims to provide answers to the global labor market challenges of our time. Our key objective is to build bridges between academic research, policymakers and society.

IZA Discussion Papers often represent preliminary work and are circulated to encourage discussion. Citation of such a paper should account for its provisional character. A revised version may be available directly from the author.

	IZA – Institute of Labor Economics	
Schaumburg-Lippe-Straße 5–9 53113 Bonn, Germany	Phone: +49-228-3894-0 Email: publications@iza.org	www.iza.org

ABSTRACT

Heterogeneous Layoff Effects of the US Short-Time Compensation Program^{*}

The Short-Time Compensation (STC) program enables US firms to reduce work hours via pro-rated Unemployment Insurance (UI) benefits, rather than relying on layoffs as a costcutting tool. Despite the program's potential to preclude skill loss and rehiring/ retraining costs, firms' participation rates are still very low in response to economic downturns. Using firm-level UI administrative data, we show why by illustrating which type firms benefit from the program and which do not. Semiparametric estimation indicates STC reduces layoff rates for cyclically sensitive firms by about 15%, but has no effect for more cyclically stable firms.

JEL Classification:	C21, C38, J63, J65
Keywords:	short-time compensation, layoffs, inverse probability weighting,
	heterogeneity, finite mixture model

Corresponding author:

Solomon W. Polachek Department of Economics State University of New York at Binghamton Binghamton, NY 13902 USA E-mail: polachek@binghamton.edu

^{*} The authors are grateful to Alfonso Flores-Lagunes, Daniel Hammermesh, Christopher Hanes, Chanita Holmes, Chinhui Juhn, and Bong Joon Yoon for their constructive comments and suggestions which substantially improved the paper.

1 Introduction

The US Short-Time Compensation (STC) program provides pro-rated benefits to employees who are forced to work reduced hours when demand temporarily slackens.¹ Benefits are paid from the state Unemployment Insurance (UI) trust fund, which are financed via the employer's payroll taxes. Firms taking advantage of this program are expected to retain potentially valuable workers, and as a result, local economies need not suffer from higher unemployment rates.² In light of the Great Recession in 2009, there has been a resurgence of interest in STC, with an additional 11 states enacting STC provisions, bringing the current total to 28 states. Yet, the participation rate of firms is still very low because they continue to rely on layoffs as a cost-cutting tool. We assess whether and to what extent the STC program is an effective alternative to layoffs by US firms.

Empirical support for STC as a prevention of job loss during economic downturns is mostly limited to aggregate cross-country studies (Abraham & Houseman, 1994; Boeri & Bruecker, 2011; Cahuc & Carcillo, 2011; Hijzen & Martin, 2013). From these studies, it can be hard to make conclusive statements about the success of the STC program in a given country without considering institutional differences and interactions with other labor market policies (Arico & Stein, 2012).³ Using state-level manufacturing production data for the US, Abraham & Houseman (2014b) find that manufacturers in about half of the states that used STC during the 2008-2009 recession rely more on work hour adjustment than those in non-STC states.

Micro-level studies designed to test the findings of aggregate studies are scant and inconclusive. To our knowledge, there are only two US firm-level (evaluation) studies that estimate the STC program's impact on layoffs; the rest are concentrated on German and French firms. Kerachsky et al. (1986) estimate the extent to which the US STC program protected jobs in the 1982-1983 recession. They use UI administrative and employer survey data on firms in California, Arizona, and Oregon (the first three states to implement STC programs) and adjust for observable covariates via OLS to show that STC reduces the percent of time spent on regular UI by about 12%.⁴ Needels et al. (1997) find the opposite results when they use administrative data for California, Kansas, Florida, New York and Washington and a matching technique (as well as OLS) to compare firms that use STC following the 1990s recession

¹The pro-rated benefits paid are the share of Unemployment Insurance benefits a worker receives if laid off. The share is based on the fraction of work hours lost.

²STC benefits to the firm also include the savings from the wage, rehiring and retraining costs avoided, the flexibility gained in labor input adjustments, and the avoided adverse morale and productivity problems associated with being laid off. However, it costs to use sub-optimal amounts of labor input given the lower output demand and, as required by some state programs, to maintain fringe benefits during the STC plan.

³Arico & Stein (2012) argue that the more favorable institutional features of Germany's labor market contributed to better employment outcomes than Italy during the recent crisis, despite similar well-established STC policies. The institutional features of Germany's labor market include strong cooperation between employer and union, an efficient income support system and the interaction with other job retention policies.

⁴However, when non-STC firms with no propensity to use STC are removed from the sample (10% of sampled firms), the estimated effect falls and become insignificant. Their employer survey was used to identify the non-STC employers that consciously rejected the use of STC as having no propensity to use STC.

to non-STC firms.⁵ Even firm-level studies in Germany and France provide limited guidance regarding what results to expect in the context of the US. Like Needels et al. (1997), Calavrezo et al. (2009, 2010) find that STC increases the incidence or extent of layoffs in France, even while using more rigorous IV/inverse Mills ratio techniques (Calavrezo et al., 2009) and propensity score matching (Calavrezo et al., 2010).⁶ Whereas Boeri & Bruecker (2011) employ an IV method to show that a greater share of STC workers has a positive impact on employment growth by German plants in 2008-2009, Bellman et al. (2012) and Kruppe & Scholz (2014) find no significant effect using the same data, albeit using different methods (Bellman et al.: IV difference-in-difference and Kruppe & Scholz: propensity-score matching).

The contradictory evidence from micro-level studies reflects the complex nature of the relationship between STC use and layoffs that is perhaps driven by selectivity. There are two key reasons why firms that select into STC may differ significantly from those that do not. First, firms using STC are prone to excessive layoffs in times of economic hardship, a positive selection bias since these firms are more likely to use the program. But second, a negative selection bias may occur if firms that choose to use STC prefer hours reductions for retaining valuable workers (rather than layoffs) because these firms actually have the best prospects in response to a positive demand shock. These selectivity issues have made it difficult for existing studies to ascertain the impact of STC.

Our contribution is two-fold. First, we tackle both selection problems by econometrically classifying firms based on their workforce dynamics. To do so, we employ a finite mixture model (FMM) to identify latent groups of firms with statistically similar histories. We define relatively cyclically sensitive firms as ones having a history of employment fluctuations and a reliance on layoffs or work hour reductions, perhaps because they have a lower capitalized value to sustain losses or lower adjustment costs.⁷ Also, to further guard against negative selection bias, we control for employee skill as a simulated unmeasured confounder à la Ichino et al. (2008).⁸

Second, given that no studies, to our knowledge, examine how STC's impact varies by firm characteristics, we show that the program may affect firms differently depending on their industry, labor costs, degree of workforce stability, and whether they are subsidized by the UI tax system. Thus, we identify heterogeneous layoff responses to a specific countercyclical benefit program, and thus contribute to the literature on labor hoarding in response to business cycle shocks. Accounting for firm heterogeneity is also important from a policy perspective because determining the relationship between STC effects and firm characteristics pinpoints the type of firms most likely to benefit. Knowing which firms benefit, and which

 $^{^5\}mathrm{Comparison}$ non-STC firms were selected based on firm size, UI tax rate, and industry.

 $^{^{6}}$ Calavrezo et al. (2009) and other IV studies employ experience with STC (i.e. lagged STC take-up rate) as an instrument. Such estimates may be biased if the firms with past STC experience have a history of economic instability, or it could be that their early adoption indicates better management practices (or administrative capacity) that affect their resilience to demand shocks.

⁷Technical details are provided later.

⁸We adopt these approaches because there are no obvious valid instruments triggering STC use and there are inadequate data for fixed-effects estimation. Our approach enables us to estimate heterogeneous effects of STC based on latent classes of workforce dynamics.

do not, sheds light on why many firms do not participate. But also, identifying the characteristics that make STC more effective can enable policymakers to target firms at which the program will have the greatest impact.

We seek to answer three questions. Does STC deter layoffs? If so, by how much? And third, does STC work for all firms? We semi-parametrically estimate potentially heterogeneous effects of STC using UI administrative data on 2420 firms in five US states (California, Florida, Kansas, New York, and Washington) over the 1991-1993 period.⁹ The FMM identifies latent groupings of stable and unstable firms based on cyclical sensitivity implicit in the quarterly 1991-1992 data on layoffs, work hour reductions, and employment fluctuations. Given this grouping, we reweight the 1993 subsample to account for non-random firm drop-outs and for firm selection into STC. These reweighted data are used to estimate the effect of STC on layoff rates.¹⁰ This weighting scheme better allows us to compare the mean layoff rate of STC firms to their mean layoff rate had they not used STC. Such a comparison is then made based on firm characteristics. We find that the program is not effective for the typical STC firm, but that it very effectively reduces layoff rates for unstable STC firms by about 15%. Even greater reductions in layoff rates can be achieved if firms are exempt of tax charges for STC benefits when the program is targeted to unstable firms with significant labor costs.

The remainder of the paper is structured as follows. Section 2 reviews US participation in the STC program. Section 3 makes predictions about the effect of STC and how it may vary by firm characteristics. Section 4 describes the data. Section 5 outlines our three-step empirical strategy. Section 6 reports our estimation results, which includes checks for potential violations of the conditional independence identifying assumption. The final section summarizes and makes some concluding remarks regarding policy implications.

2 STC program participation in US

California was the first state to implement an STC program in 1978, following the mass public sector layoffs resulting from proposition 13 limiting state spending (Kerachsky et al., 1986). The onset of the 1982-1983 recession further motivated compensated work-sharing, which led to an extensive adoption of similar programs by an additional 18 states until 1992. Despite Congress enacting the first permanent STC legislation in that year, program implementations were dormant until 2009 when renewed interest in STC arose. This resurgence culminated in the development of a new federal law, passed in 2012, which further supported state adoption of STC programs. Currently, a slight majority of states amended their UI laws to allow increased program participation. They vary greatly in take-up rates and program-specific rules, but despite the growth in the number of STC states, program use has generally remained low. In 1991, STC benefits were less than 1% of regular UI benefits paid out and

 $^{^{9}}$ This US firm-level UI administrative dataset is the only one available for public-use. It contains the UI and STC records for 5 states (for reasons given in section 4) during one of the longest period of jobless recovery.

¹⁰Like other studies, we also employ a propensity-score matching method as a robustness check on the findings from the propensity-weighting method.

were still about the same in 2008 (Abraham & Houseman, 2014b).¹¹ Firms in STC states also made limited use of it. In 1994, based on 17 states, firm participation in STC averaged less than 1% of all UI participating firms (Needels et al., 1997).

It is not fully known what factors account for the low take-up of STC programs by US states and the subsequent low program participation by firms, though some contributing factors are proffered by Needels et al. (1997), Abraham & Houseman (2014a) and Balducchi et al. (2015). The low STC adoption by states may reflect resistance from stakeholders (state UI agency, unions etc.), a lack of knowledge or understanding about STC due to limited state promotions, the potential threat to the UI trust fund, concerns related to program administrative costs, and also whether STC is actually needed (e.g. if most firms in a state are small or non-manufacturing firms). States with less liberal political ideologies appear less likely to implement STC (Vroman, 2013). Employers themselves may be attracted to the program for different reasons. Kerachsky et al.'s (1986) and Balducchi et al. (2015) identify industry, financial health and human capital factors (wage and tenure) as factors associated with employers' STC use. Maintaining full fringe benefit and UI tax implications can also be relevant to employers' decisions to use STC; but Kerachsky et al. (1986) find little or no support for them. All these factors may have led to differential STC take-up by US firms.

3 Predicting layoff effects of STC

It is hard to predict whether using STC is an effective alternative to layoffs. The existing empirical work reveal three possible effects. First, STC can preclude layoffs by making it more cost-effective for firms to retain workers for when output demand picks up and full labor input is required. It is cost-effective in the sense that it can save employers in wage costs and preclude rehiring costs and skill losses. Second, STC may have no effect on layoffs if the firm that uses STC would have retained workers anyway in the absence of the program. Third, STC may promote layoffs, as demonstrated in the studies on French firms. One interpretation is that the recourse to STC is "a way for establishments to calm the social tensions before a planned redundancy scheme" (Calavrezo et al., 2009, p. 13). Another interpretation is that STC is positively related to layoffs because the program is typically aimed at firms in structural decline —a positive selection bias.

To further complicate matters, it is also possible that some firms benefit while others do not. How firm-specific characteristics affect STC's impact on layoffs is an issue *not* currently addressed in the empirical literature. For instance, cyclically sensitive less stable firms, with revenues that are so sensitive to demand conditions that they have unstable employment and habitually rely on laying off workers or reducing work hours when demand is low, may benefit more than the typical firm. For these firms, we expect the impact of STC to depend on the extent of firms' workforce stability because less stable firms are more exposed to excessive layoffs.

 $^{^{11}{\}rm The~STC}$ usage rate was higher than 1% during the 2009-2010 recessionary years, but still was quite low by international standards.

Similarly, a firm's degree of experience rating can affect the program's effectiveness. Firms that incur UI tax charges that do not cover the benefits paid out to employees laid off or to those with shorter work schedules are said to be imperfectly experience rated. This is especially true for those firms at their state's statutory minimum or maximum tax rate.¹² These firms incur little or no marginal cost, and thus are effectively subsidized, since their tax rates tend not to increase with further reliance on compensated unemployment (UI and STC) benefits (Abraham & Houseman, 2014b). Thus, conditioning on firms' history of workforce dynamics, we expect that such subsidized STC firms would be more induced to make work-hour adjustments under STC instead of resorting to layoffs.¹³ In other words, if the extent to which firms share in the cost of STC provision reflects the degree of generosity of the program (Hijzen & Venn, 2011), then a more generous STC program should be more effective at reducing layoffs (Van Audenrode, 1994).

Likewise, firms with high labor costs may also react quite differently to STC use than firms with low labor costs. If indeed high-wage paying firms attract more productive workers (Abowd et al., 1999), then they are likely to suffer greater human capital losses from lay-offs, and hence higher rehiring and retraining costs. Dube et al. (2010) find evidence of a positive relationship between such replacement costs and wage. Therefore, consistent with Van Audenrode's (1994) theoretical findings, we can expect higher replacement costs faced by high-wage paying firms to foster greater substitution of employment reduction in favor of shorter work schedules.

Finally, the effect of STC may vary by the industry in which firms operate. Manufacturing firms are more likely to use STC than other firms (MaCurdy et al., 2004). Since manufacturing firms (electronics, industrial machinery etc.) tend to use specialized labor skills that are mainly acquired through apprenticeship and on-the-job-training, the effect of STC may be stronger for such firms to avoid the costs of retraining and rehiring when business improves. However, since manufacturing firms also tend to be more capital intensive (Helper et al., 2012), they might find it relatively more expensive to adopt shorter work schedules. For instance, high start-up and shutdown costs of capital equipment for some manufacturing operations may discourage frequent plant shutdowns associated with shorter work shifts (Lilien & Hall, 1986).¹⁴ By this argument, STC's impact may be moderated for manufacturing firms.

¹²In our sample, virtually all subsidized firms are at the maximum tax rate. Less than 1% of firms are at the minimum tax rate. We also considered states with special provisions that affect the maximum UI tax rate. In the 1990's, these entail only two states, Florida and Missouri, that apply a super maximum tax rate to STC employers with tax liabilities exceeding the state's maximum UI tax rate (Needels et al., 1997). So, in those states, we treat STC firms as subsidized if they are at the super maximum rate, otherwise not.

¹³Whereas typically being at the maximum UI tax rate may indicate a distressed firm's past economic difficulties, this is not the case here because we condition on workforce stability in our analyses.

¹⁴It is worth noting that, according to Mattey & Strongin (1997), just about 50% of manufacturing firms operate 24 hours per day for at least 5 days per week, based on the Census Survey of Plant Capacity microdata (1979-1988).

4 Data

We employ Needels et al.'s (1997) public-use, US firm-level, UI administrative data.¹⁵ It records 1991-1993 data for a sample of 3415 firms in California, Florida, Kansas, New York and Washington. Of the 17 states with active STC programs at the time, these 5 states were reported to have substantial employer participation, adequate and accessible records of key variables, and were diverse in terms of geography, economy, demography and program-specific rules (see table 1). Whereas the sample contains all firms in Florida, Kansas and Washington that filed an STC plan in 1992, it only includes a random sample of STC firms from California and New York.¹⁶ The sample also includes comparison firms with no STC plan over the same time period. The corresponding non-STC firms are similar to the STC firms based on Standard Industrial Classification (SIC) code, UI tax rate, and employment size.¹⁷ These data still have relevance for current analysis, given that the basic structure of the STC program and its adoption rate have remained low.

Our layoff measure is the number of new UI claimants to the firm.¹⁸ We estimate the effect of STC use on layoff rates in 1993.¹⁹ We employ firms' 1991-1992 histories to implement the FMM (described in the next section) for uncovering latent classes of firms that are statistically similar in workforce cyclicality. The FMM uses quarterly data on total compensated unemployment benefit (as a fraction of full-time equivalent payroll) and employee count (as a fraction of the maximum count over 1991-1992).²⁰ We exclude 915 firms with missing values in the variables employed in the FMM. In any case, the distribution of the layoff rate of firms operating in 1993 and the proportion that use STC (35%) remains virtually unchanged. In the remaining sample, about 3% of firms closed business in 1993. Such firm drop-outs may create a non-random selection of firms if they tend to be the economically weak ones. Our empirical strategy adjusts for firm drop-outs. The estimation sample has 2420 firms.

¹⁵Data downloaded from the Employment Research Data Center of the W.E. Upjohn Institute for Employment Research. Data on 47 firms were not published to restrict their identity. In addition, the data contain only the one-digit Standard Industrial Classification code.

¹⁶Program participation in California and New York far exceeded those in other states.

¹⁷Needels et al. (1997) choose comparison firms (those not filing an STC plan) based on three-digit SIC, quintiles of the 1992 UI tax rate, and quintiles of employment size. Firms that file STC plans are not necessarily the same ones that actually end up using STC in a given period. Our analysis is based on use of STC rather than simply filing for a plan. So, we control for industry (one-digit SIC), and for the levels (rather than quintiles) of employment and UI tax rate via a propensity score weighting procedure.

¹⁸Since UI claims are usually filed after job loss, it should trend closely with actual layoffs and discharges series. Note, however, that those employees who are eligible for UI but do not claim benefits are not accounted for by this layoff measure. Also, some UI claims drawn on a firm may not reflect layoffs by that firm if, for instance, the laid off worker had more than one base-period employer. However, we suspect that, in large part, these issues are inconsequential.

¹⁹Layoff rate is computed as the number of new UI claimants in 1993 as a proportion of employment size in 1992. We apply Vandervieren & Hubert (2004) method to identify outliers from this layoff rate variable. The method is applicable to variables that have naturally skewed distributions. Thirteen layoff rate observations are discarded as outliers.

²⁰Full-time equivalent payroll is the total wage an employer would have paid out if no workers were laid off or their hours reduced. It is computed as the total wage plus 2 times the sum of regular UI benefits and STC benefits, assuming a replacement rate of 50%. In regard to employee count, by dividing it by the maximum count over 1991-1992, we constrain it between 0 and 1 to allow for comparison across firms.

The first row of table 2 reports an average layoff rate in 1993 of 12.5%, with STC firms' layoff rate (13.8%) being higher than that of non-STC firms (11.8%). Based on the next two variables in table 2, it appears that the typical STC firm (unlike the non-STC firm) suffered a decline in workforce over the 1991-1992 period from 79 to 75 workers. This pattern is reflected in a higher rate of increase (from 2% to 3%) in unemployment benefits charged to STC firms (as a percent of payroll), relative to non-STC firms (from 1.5% to 2.0%). Although these descriptives (on annual data) indicate STC firms are larger but more cyclically sensitive (being more prone layoff workers), we make a more rigorous assessment using the FMM at a later stage.

The remaining variables in table 2 may affect selection into STC. While STC firms tend to be bigger in terms of employee count than non-STC firms (over 70 vs. about 50 workers), the average (log) wage per employee is not significantly different. Regarding firms' tax contribution, we normalize UI tax rates (at the start of 1993) to lie between -1 and 1, for comparability across states, since each US state sets different statutory minimum and maximum tax rates.²¹ A value of -1 or 1 means that a firm is paying the minimum or maximum tax rate, respectively. Thus, UI benefits charged to that firm do not affect its tax payments and the firm is said to receive a pure subsidy (i.e. imperfectly experience rated). Table 2 shows that STC firms incur a higher normalized tax rate (0.54 vs. 0.38). They are also more likely to receive a pure subsidy (31.5% vs. 20.9%). The table also shows that the total tax contribution of firms to the UI trust fund over 1991-1992, on average, do not cover the benefits paid out. That is, the net balance (as a percent of taxable wage) is -5.1%, and it gets more negative for STC firms (-8.0%). This means tax rates do not rise sufficiently to cover the full UI benefits, and more so for STC firms.

The sample covers the industrial sector and US states where STC use is most common. Manufacturing firms are known to be popular users of STC (MaCurdy et al., 2004). So, it is encouraging that they are well-represented in the overall sample (45.1%) and also more representative of the sample STC firms (50.2%) than the non-STC ones (42.4%). As expected, California and New York comprise the majority of our sample (64.3%), and both represent the majority of STC firms in the early 1990's.²²

5 Empirical strategy

5.1 Parameter of interest

The effect of firms *using* STC on their layoff rates is measured by comparing the layoff rates of these STC firms to their layoff rates had they not used STC. Formally, let S be an

²¹Let t_{is} represents the UI tax rate (at the start of 1993) for firm *i* in state *s*. Then, we normalize the UI tax rate as follows for firm *i* in state *s*: $[2t_{is} - \max(t_{is}) - \min(t_{is})]/[\max(t_{is}) - \min(t_{is})]$. Our basic results are not sensitive to this normalization; they are qualitatively unchanged when the untransformed tax rate is used.

 $^{^{22}\}mathrm{We}$ account for state fixed effects to capture differences in economic conditions, UI laws and labor market conditions.

indicator variable, which takes on 1 if a firm uses STC and 0 if it does not. If the firm uses STC (S = 1), its layoff rate is ℓ_1 , and it is ℓ_0 if that same firm does not use STC (S = 0).²³ We estimate the average effect of firms' STC use on their layoff rates, which varies based on firm-specific attributes. Our parameter of interest is

$$\beta(\boldsymbol{z}) = E(\ell_1 | S = 1, \boldsymbol{Z} = \boldsymbol{z}) / E(\ell_0 | S = 1, \boldsymbol{Z} = \boldsymbol{z}) = \beta_1(\boldsymbol{z}) / \beta_0(\boldsymbol{z})$$
(1)

where \mathbf{Z} is a vector of firm characteristics that impact the effectiveness of the STC program (e.g. labor costs); $\beta_1(\mathbf{z})$ is the mean layoff rate among STC firms with particular characteristics \mathbf{z} ; and $\beta_0(\mathbf{z})$ is their counterfactual mean layoff rate had the firms not used STC. We interpret the quantity $[\beta(\mathbf{z}) - 1]$ as the relative change in the layoff rate of STC firms with particular characteristics \mathbf{z} .²⁴ Let β be the relative STC effect that is unconditional on \mathbf{Z} . We follow a three-step procedure to estimate the parameter $\beta(\mathbf{z})$ in equation 1:

- a. Distinguish firms based on their history of workforce dynamics using the FMM.
- b. Estimate the constant relative STC effect β , using two probability weights to:
 - (i) adjust for firm drop-outs by up-weighting firms in the estimation sample that are similar in workforce dynamics; and
 - (ii) adjust for selection into STC by up-weighting non-STC firms closest to the STC ones (in terms of observables) among firms with similar workforce dynamics.
- c. Non-parametrically estimate β for a given firm with $\mathbf{Z} = \mathbf{z}$, that is $\beta(\mathbf{z})$, where \mathbf{Z} also includes an indicator for firms' degree of workforce stability.

5.2 Identifying firms differing in workforce dynamics

In this study, a cyclically unstable firm is defined as one which has a history of high reliance on work hour reductions, layoffs, and unstable employment during or following an economic downturn. Such unstable firms often react by laying off employees or reducing unit labor costs to raise liquidity in order to pay current debts (John et al., 1992; DeAngelo & DeAngelo, 1991). Based on firms' histories of compensated unemployment charges and employment adjustments, we employ FMM to create different groups of firms that vary in workforce stability. The FMM generally identifies latent groups of firms, given that a priori firms differ by their class of cyclical workforce dynamics, which we denote as C.

To describe the FMM procedure, consider the multivariate vector $\boldsymbol{y}_{i,1:T} = (\boldsymbol{y}'_{i,1} \dots \boldsymbol{y}'_{i,T})'$, which in our data represent the history of total compensated unemployment charges and

²³Like Kerachsky et al. (1986) and Needels et al. (1997), we focus on participation of firms in the STC program (extensive margin) rather than firms' STC usage rate (intensive margin) adopted by non-US studies. We believe, in light of the low prevalence of STC firms in the US, the participation decision is the more important margin on which to assess the success of the STC program in reducing firms' layoff rate.

²⁴We can also interpret $\beta(z)$ as a layoff conversion rate —the degree to which an STC firm's full-time equivalent workers substitute for the layoffs it would have otherwise experienced. $\beta(z) = 0$ implies a perfect substitution so that each full-time equivalent worker on STC replaces a laid off worker in firms with attributes z.

employment adjustments over T quarterly time periods for firm i. We might expect for instance that $\mathbf{y}_{i,1:T}$ for relatively cyclically unstable firms is distributed differently from relatively stable firms. If so, then "the observed data [on $\mathbf{y}_{i,1:T}$] are drawn from a mix of distinct underlying populations" (Greene, 2012, p. 630). One can define the probability (π_g) of a firm being drawn from one population group g or the other and write out the density $f(\cdot)$ for $\mathbf{y}_{i,1:T}$ as a weighted sum as follows:

$$f(\boldsymbol{y}_{i,1:T}; \boldsymbol{\theta}_{1:G,1:T}, \pi_{1:G}) = \sum_{c=1}^{G} \pi_c \cdot \boldsymbol{\phi}(\boldsymbol{y}_{i,1:T}; \boldsymbol{\theta}_{c,1:T})$$
(2)

where G represents the number of groups, 1:G denotes parameters for groups 1 to G; and $\phi(y_{i,1:T}; \theta_{c,1:T})$ represents a conditional (on group c) multivariate normal density of $y_{i,1:T}$ for firm i with group-specific parameters that vary across time contained in $\theta_{c,1:T}$. The parameter $\theta_{c,1:T}$ (especially the mean) is useful for describing the differences in firm groupings in terms of the extent of reliance on layoffs and work hour reduction and employment adjustments over time. The number of groups (G) and the parameters ($\theta_{1:G,1:T}, \pi_{1:G}$) are determined entirely from the data by maximizing the log-likelihood function of equation 2.²⁵ In so doing, we avoid biases that may arise when firms are grouped by imposing arbitrary cutoff levels for multiple variables in $y_{i,1:T}$. The population group from which an observed firm is drawn can also be determined by using equation 2 and Bayes' theorem.²⁶ The idea is to assign a firm with a given history $y_{i,1:T}$ to the group c if $y_{i,1:T}$ most likely derives from a density ϕ with parameter $\boldsymbol{\theta}_{c,1:T}$. The result is an estimated ordered categorical variable \hat{C} that classifies firms with similar workforce histories. We then account for \hat{C} in two ways. First, it is used to achieve greater similarity between STC and non-STC firms via probability weights for dealing with selectivity issues (step b). Second, it is included in Z to estimate STC effects that also vary by the extent of workforce stability (step c).

5.3 Estimate constant STC effect

To estimate β , we address two selectivity issues: sample attrition (drop-outs) and selection into STC. First, we consider firm drop-outs. If in the data, some cease operating during or following a downturn, then such firms are more likely to be economically weak and the observed data is likely to contain a non-random selection of firms. To make our sample representative of the contribution of such weak firms that later drop-out, we up-weight observed firms that are similarly unstable. To do this, we estimate the probability of not dropping out, given firms' workforce histories $Pr(A = 0|\hat{C})$, where A indicates whether a

²⁵The log-likelihood function is maximized (using an E-M algorithm) first for a single group of firms and then for 2, 3, 4 and 5 groups of firms. Fraley & Raftery (2002) suggest that the number of groups be small to avoid groups containing a few observations that may cause the E-M algorithm to fail or give erroneous results. They recommend the optimal value of G that minimizes the Bayesian Information Criterion.

²⁶Formally, a firm *i* with a given history $y_{i,1:T}$ belongs to group *c* if it maximizes the conditional probability of belonging to that group, $\pi_{c|y_{i,1:T}} = \pi_c \cdot \phi(\mathbf{y}_{i,1:T}; \boldsymbol{\theta}_{c,1:T}) / f(\mathbf{y}_{i,1:T}; \boldsymbol{\theta}_{1:G,1:T}, \pi_{1:G})$. For example, if there are two groups (unstable and stable firms), a firm is classified as unstable if the conditional probability of being unstable is larger than 0.5.

firm later drops out.²⁷ We then define a variable $w = 1/Pr(A = 0|\hat{C})$, which assigns a weight to observed firms that corrects the number of firms represented in each category of workforce stability (\hat{C}) . When w is applied, less more cyclically sensitive weaker firms are given more weight.

Second, we consider firm selection into STC. The problem here is that one cannot observe the mean layoff rate for STC firms had they not used STC, that is, $E(\ell_0|S = 1)$. If firms that use STC are systematically different from firms that do not, then one cannot simply set the observed mean layoff rate for non-STC firms $E(\ell_0|S = 0)$ equal to $E(\ell_0|S = 1)$. Let \boldsymbol{X} be observed firm characteristics that affect the decision to use STC. Our key identifying assumption is that the probability of selection into STC depends on \boldsymbol{X} within each estimated category (\hat{c}) of employment history, that is, $p_{\hat{c}} = Pr(S = 1|\boldsymbol{X}, \hat{C} = \hat{c})$. As such, conditioning on $p_{\hat{c}}$ is enough to ensure the layoff rate in the absence of using STC (ℓ_0) is not different for STC and non-STC firms. In other words, conditional independence of ℓ_0 and S is assumed to be achieved within each latent class of employment history.²⁸

In estimating $E(\ell_0|S=1)$, we weight non-STC firms' layoff rates by $w_{\hat{c}} = p_{\hat{c}}/(1-p_{\hat{c}})$, where $p_{\hat{c}} < 1$. More weight is assigned to non-STC firms with relatively high odds $w_{\hat{c}}$ of using STC based on their characteristics (\hat{C}, \mathbf{X}') but still choosing not to use the STC program. We consider such firms similar to STC firms. We apply inverse probability weighting (IPW) to estimate the mean counterfactual outcome $E(\ell_0|S=1)$ as $E(w_{\hat{c}}(1-S)\ell)/E(S)$, where ℓ is the observed layoff rate (see Wooldridge, 2002, p. 615).²⁹ The mean layoff rate of STC firms $E(\ell_1|S=1)$ is easily computed as $E(S\ell)/E(S)$. Let \hat{w}_i and \hat{w}_{i,\hat{c}_i} be the estimated attrition and selection weights respectively for firm *i*. Since $\beta = E(\ell_1|S=1)/E(\ell_0|S=1)$, using the sample counterpart of the expectation operator, we estimate the constant relative STC effect corrected for attrition and self-selection bias as follows:

$$\hat{\beta} = \frac{\sum_{i=1}^{n} \hat{w}_i S_i \ell_i}{\sum_{i=1}^{n} \hat{w}_i \hat{w}_{i,\hat{c}_i} (1 - S_i) \ell_i}$$
(3)

Basu et al. (2008) provide support for the consistency of the IPW for response variables that are non-negative and positively skewed (e.g. layoff rate).³⁰ The IPW estimator is also competitive with matching estimators when there are sufficient comparable treated and control units (Busso et al., 2014), as in our case. Nevertheless, we use matching (via a local linear ridge regression) to support the IPW since Frolich (2004) finds it achieves the lowest mean squared error of all considered estimators (including IPW).³¹

²⁷We also include all our other observables and obtain very similar results.

²⁸Section 6.3.2 provides indirect evidence supporting the robustness of our estimates to a potentially key unmeasured omitted variables namely employee skill.

²⁹We also consider an estimator with normalized weights that sum to one and find a similar estimate of -0.095 at the 5% significant level (compared to the -0.107 reported in table 4, column 5).

³⁰Basu et al. (2008) find that the IPW estimator is always consistent when the probability model of selection is over-specified with quadratic and interaction terms as well as the main effects of X.

³¹In local linear ridge matching estimator, the counterfactual layoff rate of STC firms is estimated by a local linear ridge regression. Essentially, a ridge term (pre-chosen constant) is added to the denominator of an OLS estimator to avoid division by values close to zero. The estimator tends to improve in variance, although attenuated toward zero. Another disadvantage of this estimator is that a bandwidth parameter needs to be determined.

5.4 Estimate heterogeneous STC effects

In line with our expectations from section 3 that the effectiveness of STC depends on firm attributes (\mathbf{Z}), we now estimate β for a given firm with $\mathbf{Z} = \mathbf{z}$, that is, $\beta(\mathbf{z})$.³² Rather than commit to a particular functional form, we non-parametrically derive firm-specific estimates corresponding to each observed combination of attributes in \mathbf{Z} by modifying equation 3 as:

$$\hat{\beta}(\boldsymbol{z}) = \frac{\sum_{i=1}^{n} K_h(\boldsymbol{Z}_i - \boldsymbol{z}) \hat{w}_i S_i \ell_i}{\sum_{i=1}^{n} K_h(\boldsymbol{Z}_i - \boldsymbol{z}) \hat{w}_i \hat{w}_{i,\hat{c}_i} (1 - S_i) \ell_i}$$
(4)

where $K_h(\mathbf{Z}_i - \mathbf{z})$ is a kernel weight function that measures the distance between \mathbf{Z}_i and \mathbf{z} , and then assigns a relatively high weight if firm *i* possesses attributes similar to the combination in \mathbf{z} .³³ The parameter *h* (the bandwidth) determines the set of firms with features considered close to \mathbf{z} .³⁴ We use Li & Racine's (2010, p. 1617) specification test to check the relevance of the \mathbf{Z} variables as moderators of the effect of firms' STC use on the layoff rate.³⁵

To assess the statistical significance of $\hat{\beta}(\boldsymbol{z})$, as well as the constant effect $\hat{\beta}$, we compute standard errors using pairs bootstrap samples. Since group membership based on workforce dynamics and the probability of sample attrition and selection into STC are estimated beforehand, the sampling error inherent in their estimation must be incorporated in the standard errors of $\hat{\beta}$ and $\hat{\beta}(\boldsymbol{z})$. As such, the entire three-step procedure, just outlined, is bootstrapped 500 times to compute standard errors.

6 Estimation results

At the outset, we identify the latent groups of firms that differ in employment histories. Conditional on these groups, we then compute two probability weights for sample attrition and selection into STC. These weights are used in estimating the effectiveness of the STC program in precluding firms' layoffs. We then check for violations of our identifying assumption.

³²Under our identifying assumption, when \boldsymbol{Z} is a subset of $(\hat{C}, \boldsymbol{X}')$, it can be shown using the tower property that $E(\ell_1|S=1, \boldsymbol{Z}) = E(S\ell|\boldsymbol{Z})/E(S|\boldsymbol{Z})$ and $E(\ell_0|S=1, \boldsymbol{Z}) = E(\hat{w}_{i,\hat{c}_i}(1-S)\ell|\boldsymbol{Z})/E(S|\boldsymbol{Z})$.

³³As in Li & Racine (2007), we used a generalized product kernel function $K_h(\mathbf{Z}_i - \mathbf{z})$, comprising of a mixture of second-order Epanechnikov (for continuous data) and Aitchison-Aitken (for binary data) kernels, to weight neighborhood observations such that weights increase with closer proximity to \mathbf{z} .

³⁴The bandwidth h is chosen by cross-validation (CV). The sum of squared out-of-sample forecast error of STC firms' layoff rate is minimized: $h^{CV} = \arg \max_h \sum_{i:S_i=1} [\ell_i - \hat{\beta}_1^{-i}(\boldsymbol{z}_i)]^2$, where $\hat{\beta}_1^{-i}(\boldsymbol{z}_i)$ is the predicted mean layoff rate of STC firm i with attribute \boldsymbol{z} after leaving it out of the sample.

³⁵We use Li & Racine's (2010) test statistic to test the null hypothesis: $\beta_1(\boldsymbol{z}_i) = \beta_1$. Under the null, $\ell_i^* = \beta_1 x_i^* + \epsilon_i^*$, for $i: S_i = 1$, where $\ell_i^* = \sqrt{\hat{w}_i} \ell_i$, $x_i^* = \sqrt{\hat{w}_i}$ and $\epsilon_i^* = \sqrt{\hat{w}_i} \epsilon_i$ is an error term.

6.1 Latent firm groupings

Using quarterly 1991-1992 data on both the firms in the estimation sample and those that dropped out in 1993, we find the FMM results suggest the presence of two groups of firms with different workforce dynamics. This number of groups is optimal in the sense that it corresponds to the lowest Bayesian Information Criterion value. In other words, the gain in goodness-of-fit from segmenting the data into two groups of firms is 18%, after accounting for the loss in degrees of freedom from estimating the FMM parameters.

Figure 1 shows how the groups differ in terms of their histories of work hour reductions and layoffs and their instability in employment over the 1991-1992 period. The solid (dotted) line indicates the mean historical trajectory for firms in group 1 (group 2), as estimated from the FMM. Based on firm charges for compensated unemployment benefits (as a fraction of fulltime equivalent payroll), figure 1a shows that firms in group 1 have a history of high reliance on work hour reductions and layoffs. Although for both groups unemployment worsened somewhat following the end of the 1990-1991 downturn, it is clear that firms in group 1 are more sensitive to demand fluctuations. We also use data on employee count as a proportion of the maximum count attained over the 1991-1992 period. A lower proportion means a greater deviation in employee count from the maximum level attained over the period. Figure 1b shows that firms in group 2 are quite stable as they generally hovered around a proportion of 0.85 employed. However, the firms in group 1 appear to suffer greater instability in their employment adjustments to demand fluctuations. Their proportions are well below 0.85 especially in the first quarter of 1992 when US employment reached its trough. We refer to firms in group 1 (group 2) as cyclically unstable (stable) firms, which consists of about 39%(61%) of the firms in the sample.

The grouping of firms by extent of workforce stability via the FMM is consistent with expected patterns.³⁶ First, STC firms are more likely to be cyclically unstable. The odds of a cyclically unstable firm using STC is nearly two times higher than that of a cyclically stable firm; hence, the importance of controlling for extent of workforce stability. Second, those firms that dropped out of the sample in 1993 are more likely to be unstable. The odds are about four times greater that a cyclically unstable firm will not remain in business than a cyclically stable firm. This finding suggests that there may be a non-random selection of firms in our estimation sample. As such, the contributions of unstable firms in the estimation sample are given a higher weight to address bias due to sample attrition.

6.2 Firm selection into STC: overlap and balance

To estimate the mean layoff rate for STC firms had they not used STC, we give higher weights to non-STC firms that have high odds of selecting into STC. As such, the prob-

³⁶We also relate our grouping of firms to other firm characteristics available from a subsample of 500 firms (surveyed by Needels et al. (1997) that used STC in 1992. As expected, unstable firms are more likely to have characteristics indicating financial constraints, such as no offer of health benefits or retirement plan and experiencing a financial loss. Characteristics we expect to be unrelated to a firm's workforce stability, such as the percentage of employees of a given gender or race, are indeed unrelated.

ability of selecting into STC is estimated as a function of the covariates listed in table 3, their quadratics, and two-way interactions of the workforce stability indicator with all other covariates. In line with the suggestions of Brookhart et al. (2006), all covariates are either related to the layoff rate or related to both the layoff rate and whether firms use STC. The use of the quadratic and interaction terms is consistent with the approach of Dehejia & Wahba (1999), Morgan & Todd (2008) and Basu et al. (2008).³⁷

We include in further analysis only STC and non-STC firms that have comparable values of the estimated probability of selection into STC (the overlap or common support region). This approach ensures the availability of both STC and non-STC firms that have similar propensity to use STC. As suggested by Imbens (2004), we detect the extent of overlap by plotting separately for STC and non-STC firms the density of the estimated probabilities of selecting into STC. From figure 2, the two densities overlap quite well. In only about 3% of the cases, there are no estimated probabilities in common to STC and non-STC firms in the tails of the distributions. Thus, the estimation sample comprises the common support region (2353 observations).

We verify that the probability weights for selection into STC increase the similarity of STC and non-STC firms by performing a balancing check. As is typical in the literature (see for e.g. Morgan & Todd (2008)), we use the absolute standardized difference (ASDIFF) to compare the mean of a given covariate for STC firms to its weighted mean for non-STC firms. Table 3 provides the ASDIFF values and the means of the covariates separately for STC and non-STC firms, when non-STC firms are weighted (in the common support region) and also when they are not weighted (without the common support restriction). It is clear that the estimated probability weights substantially increase the comparability of STC and non-STC firms.³⁸ The average ASDIFF value dropped by over 90% from 0.151. In all cases, the differences in the means of the covariates are not significant after applying the weights.

6.3 Is the STC program effective?

6.3.1 Constant effect

Table 4 reports the estimates of the constant effect (β) of firms using STC on their layoff rates in 1993. Column 1 provides a simple difference-in-means estimate that compares the mean layoff rate of STC firms to that of non-STC firms. Using STC significantly (at the 1% level) increases layoff rates by 16.5%. However, when firms' degree of workforce stability is controlled for in column 2, the effect falls by about 12 percentage points and is not significantly different from zero. Further controlling for other covariates (including quadratics

³⁷We also considered cubic variables and other two-way interactions, and estimated the probability of STC use by a local linear regression (instead of a probit regression), but covariate balance deteriorated. This is also true if we directly include the quarterly history of total compensated unemployment charges and employment adjustments in the propensity score.

³⁸Since the estimated workforce stability indicator is a latent variable that groups firms that have a similar history of workforce adjustment, it may also be useful in controlling for unobserved time-varying and time-invariant factors that are correlated with it.

and two-way interactions of the workforce stability indicator with all covariates), column 3 shows STC use reduces layoff rates by on average 8.2%, albeit statistically insignificant.

It is quite possible, however, that there are comparability issues from using a standard multiple linear regression. Such a regression is based on the equal weighting of observations; no common support region (allowing for extrapolation and reduced precision); and the functional form assumption of a linear effect for all covariates. We therefore use two semi-parametric methods: the matching estimator supported by Frolich (2004) and the IPW estimator (equation 3). These methods give more weight to observations that are similar, use a common support region, and make no functional form assumptions aside from estimating the probability of selection using a probit model. Based on the matching estimator. we find that on average the layoff rates of STC firms are significantly (at the 10% level) lower by 9.4% (column 4 of table 4) than if they had not used STC. In comparison, the IPW estimate is -10.7% and statistically significant at the 5% level (column 5 of table 4). As expected, these semi-parametric estimators also achieve lower standard errors than OLS. Frolich (2004) prefers the matching estimator because of the greater precision it achieves. Indeed, the standard error of the estimate based on matching (0.049) is not different from that of the IPW estimate (0.050). Given the same precision of the IPW and matching estimators in our application, and given that the former performs well when overlap is good (Busso et al., 2014), we rely on the IPW estimator for further analysis.

6.3.2 A sensitivity check

So far, we have assumed the covariates (including the workforce stability indicator) are sufficient to identify comparable non-STC firms. Here we check the sensitivity of our estimates to a possible omitted variable namely worker skills, which could drive our estimates toward zero. Suppose firms' stock of human capital (e.g. tenure, productivity), which we will simply call 'skill', affects both the selection into STC and the layoff rate, and hence threaten the validity of our IPW estimate in table 4.³⁹ Suppose high-skilled firms are more likely to use STC (MaCurdy et al., 2004) to avoid skill loss and, in the absence of the STC program, they have lower layoff rates.⁴⁰ Following Ichino et al. (2008), we check our estimate's sensitivity to such an omitted variable by simulating it and then using it in equation 3.⁴¹ That is, we

³⁹For simplicity, we use the constant IPW estimate, rather than the heterogeneous IPW estimates, in the implementation of the simulation exercise we perform next.

⁴⁰A countervailing possibility is that high-skilled workers may not favor STC because the pay reductions from participating in the program can be avoided by finding alternative full-time employment (Gray, 1998). ⁴¹Denote the unobserved confounder 'skill' as a binary variable U. Let the probability that U = 1 by STC status and layoff rate be denoted as $p_{ij} = Pr(U = 1|S = i, Y = j)$, with i, j = 0, 1 and Y = 1 if layoff rate is higher than its mean and 0 otherwise. Consequently, the fraction of firms with U = 1 by STC status is $p_{i.} = \sum_{j=0}^{1} p_{ij} \cdot Pr(Y = j|S = i)$. Note that the prevalence of firms in our sample with average wage per worker higher than the mean is about 40%. So, we assume, as in Ichino et al. (2008), that Pr(U = 1) = 0.4(the prevalence of high-skilled firms) and $p_{11} - p_{10} = 0$ (the layoff effect of U when STC is used). Then, the layoff effect of U in the absence of the program ($d = p_{01} - p_{00}$) and the effect of U on selection into STC ($s = p_1 - p_0$.) uniquely determine the parameters p_{ij} for imputing the values of U. A hundred simulations of U are performed for a given set of parameters p_{ij} , then $\hat{\beta}$ is computed for each simulation, and the mean of the estimates reported.

construct the distribution of skill so that it has a positive effect (of size s) on selection into STC but a negative effect (of size d) on the layoff rate in the program's absence. We then determine the values of d and s necessary to drive the IPW estimate close to zero. If those d and s values are implausibly large, this analysis would support the validity of the IPW estimate.

In table 5, we show how the IPW estimate varies in response to different values of d and s. When d = -0.20 and s = 0.20, the IPW estimate becomes -2.2%. To put this finding into perspective, suppose we calibrate the distribution of skill to mimic that of the average wage per worker, since high-wage firms attract more productive workers (Abowd et al., 1999).⁴² In that case, d = -0.09, s = 0.01, and the IPW estimate is -10.4% (not shown in table), which suggests a negligible effect on the baseline IPW estimate. It follows that the selection and layoff effects of skill would need to be substantially larger (in absolute value) than d = -0.09and s = 0.01 to drive the IPW estimate to zero. Even if we increase s to 0.10, the estimate still remains economically significant (about -8.3%). These findings give credence to the robustness of our IPW estimator.

6.3.3 Heterogeneous effects

In section 3, we predicted the effect of STC may vary with specific firm attributes, that is $\beta(\mathbf{z})$. The firm attributes we consider are the degree of workforce stability, whether firms operate in the manufacturing industry, extent to which firms are subsidized by the UI tax system, and wage level. Yet the methods so far presuppose the effect is the same irrespective of these features. Indeed, the model specification test of Li & Racine (2010) rejects a constant effect in favor of heterogeneous effects based on a zero bootstrapped pvalue. This test result supports the joint relevance of the four firm attributes as moderators of the effect of STC.⁴³ As such, further results are based on the estimation of equation 4, which provides firm-specific parameter estimates that correspond to each firm's combination on the four attributes of interest. We present the distribution of these estimates graphically in figure 3 and compress the results by reporting the estimates at the mean and the quartiles (P₂₅, P₅₀, and P₇₅) in panel 1, column 1 of table 6. At a later stage, we show how these results differ by specific features of firms in order to investigate heterogeneous layoff effects.

In support of the model specification test, figure 3 shows a double-peaked distribution of STC effects that clearly varies based on the features of firms. The estimates with the highest frequency occurs at about -8%, while those with the second highest frequency occurs at about -25%. The middle 50% of the estimates in the figure ranges from about -14.4% to 4.7% (statistically insignificant), with an insignificant mean estimate of -5.5% (see panel 1,

⁴²For the unobserved confounder 'skill' (denoted as the binary variable U) to mimic the distribution of average wage per worker, the latter must be transformed to a binary variable. This is done by assigning a value of 1 to firms with average wage per worker higher than its mean, and a value of 0 otherwise.

⁴³Moreover, the relatively small bandwidths associated with the Z variables (logarithm of average wage per worker, a manufacturing dummy, a workforce stability dummy and the tax rate (normalized)) of 0.424, 0.500, 0.057 and 0.198, respectively, suggest that these variables interact non-linearly with the STC effect. Hall et al. (2007) show that a bandwidth less than two times the standard deviation for a continuous variable or less than unity for a discrete variable is considered small.

column 1 of table 6). Allowing for heterogeneity, we now conclude that the typical STC firm may not benefit from STC use; but, depending on some specific features of the STC firm, its layoff rate can be reduced by over 14%. In other words, ignoring heterogeneity (as is the case in all current studies) produces a biased estimate (-10.7% versus -5.5%).

There are distinct differences in both the mean and distributional effects of STC when firms differ in workforce stability or by whether they receive a pure subsidy, but they do not differ by industry. If we distinguish firms based on their degree of workforce stability, column 1 of table 6 reveals that the STC program is only effective for cyclically unstable STC firms (panel 3). On average, the STC effect for cyclically stable firms (panel 2) is zero, but it is about 14% for cyclically unstable firms. Moreover, the distribution of estimates for unstable firms (with middle half ranging from -26.4% to -1.9%) is shifted to the left of the distribution for cyclically stable firms (ranging from -8.2% to 7.3%). In regard to industry, columns 2 and 3 show that manufacturing firms, despite being popular STC users, do not seem to benefit more from STC than firms in other industries, whether firms are cyclically unstable or stable. For clarity, we present figure 4 to show that the distribution of STC effects for manufacturing firms is hardly distinguishable from that for non-manufacturing firms. Our findings also suggest that STC has the greatest effect on cyclically unstable firms that are subsidized (from being imperfectly experience rated). Among cyclically unstable firms, STC firms that receive a subsidy reduce their layoff rates by nearly 25% on average (panel 3, column 4), whereas the mean effect is statistically insignificant for those not subsidized. Also, the middle half of the distribution of STC effects for subsidized unstable firms narrowly range from -28% to -22.7%. So, in keeping with Van Audenrode (1994), the STC program's success depends on its generosity, that is, the extent firms share in the cost of its provision.

Earlier we predicted that high-wage firms are likely to incur greater human capital losses and as such are more inclined to make work-hour adjustments to retain valuable workers. Since the STC program works best for subsidized unstable firms, we analyze the role of wages on the effectiveness of STC among these firms. Figure 5 clearly shows a positive association between wage per worker and the extent to which STC averts layoffs. At low wage levels, STC effects tend to be insignificant (shown by the gray plots). However, when wages are in the mid to high range, STC can reduce the layoff rate by 20-35%. Thus, the STC program is most effective when subsidized unstable firms incur significant labor costs. Indeed, such asymmetric benefits may account for why take-up rates are lower than many expect thus solving a paradox in the current literature.

7 Conclusion

This paper estimates the impact of the US Short-Time Compensation (STC) program on the layoff rates of STC firms. Theoretical predictions and the dearth of extant research suggest that the effect is indeterminate and potentially heterogeneous. To address selectivity issues, we compute probability weights for sample attrition and selection into STC, both of which depend on firms' cyclical sensitivities as determined from a finite mixture model. We also check for robustness to unobserved factors that could drive our estimates to zero.

If we presuppose a constant STC effect, our estimate is just about -10%. This estimate is close to the early US evaluation study of Kerachsky et al. (1986). However, that study does not account for potentially important heterogeneous effects, not examined in other studies. We predicted that the effect of STC varies by the extent of cyclical workforce stability, industry, labor costs (a proxy for human capital losses to the firm) and whether firms receive a pure subsidy (i.e. imperfectly experience-rated). Allowing for this heterogeneity, as supported by a model specification test, indicates the cyclically unstable (and not the typical) STC firm benefits from about a 15% reduction in its layoff rate. Moreover, the STC program has the greatest effect on layoff rates among cyclically unstable firms that receive a pure subsidy and potentially face significant human capital losses from layoffs. For cyclically stable firms, we find that the program has no effect, which implies that jobs covered under STC would have been retained even in the absence of the program. Interestingly, we also find that the use of STC by manufacturing firms is no more effective at reducing layoffs than in other firms. Most importantly, the heterogeneous effects we observe could explain the scant take-up rates for the program.

If the only concern is layoff unemployment, our study provides conditional support for the newer compensated work-sharing provisions passed in February 2012 to expand the STC program. This provision, in part, allows for full federal financing of STC benefits for up to three years in states that have an approved STC program. These states can then choose to relieve firms of tax charges for STC benefits paid out (Abraham & Houseman, 2014b). Our findings suggest that such subsidies to cyclically unstable firms with high labor costs (or human capital losses), in the face of relatively low demand, can yield the greatest unemployment reduction benefits.

			Maintain		Exclude
	Min. number of	% reduction in	Fringe	Duration of	part-time or
\mathbf{State}	workers affected	hours allowed	Benefits	benefits (weeks)	Seasonal workers
California	$10\%^{a}$	≥ 10	No	No limit ^b	No
Florida	$10\%^{a} \text{ or } \geq 2$	10 to 40	No	26	\mathbf{Yes}
Kansas	$10\%^{a} \text{ or } \geq 2$	20 to 40	\mathbf{Yes}	26	Yes
New York	5	20 to 60	\mathbf{Yes}	20	No
Washington	No Min	10 to 50	\mathbf{Yes}	26	Yes
Notes: ^a Proportio	n of employment at firm	or affected unit. ^b 1	No limit on we	eks, but total paid can	ot exceed 26 x weekly

Table 1: STC program features by sample states in early 1990's

benefit amount. Source: Needels et al. (1997).

	All I	Firms	STC	Non-STC	Diff
Variables	Mean	\mathbf{SD}	Mean	Mean	P-value
Lavoff rate in 1993 (%)	12.5	15.3	13.8	11.8	0.003
Compensated unempl. charges $(\%)^{a}$					0.000
in 1991	1.7	2.2	2.0	1.5	0.000
in 1992	2.4	3.1	3.0	2.0	0.000
No. of employees per firm					
in 1991	61.8	106.7	78.8	52.8	0.000
in 1992	60.5	116.9	74.7	53.1	0.000
Log. avg. wage per worker in 1992	10.1	0.5	10.2	10.1	0.366
UI Tax rate (normalized)	0.44	0.49	0.54	0.38	0.000
Purely subsidized $(\%)^{b}$	24.5	43.0	31.5	20.9	0.000
Net balance $(1991 - 1992)$ (%) ^c	-5.1	11.3	-8.0	-3.6	0.000
Industry (%)					
Mining/construction/agriculture	9.0	28.6	9.8	8.6	0.318
Manufacturing	45.1	49.8	50.2	42.4	0.000
Wholesale trade	8.6	28.0	7.1	9.4	0.050
Retail trade	7.6	26.5	6.4	8.2	0.123
Services	23.5	42.4	22.6	23.9	0.467
Other	6.2	24.2	3.8	7.5	0.000
State (%)					
California	29.1	45.4	27.3	30.1	0.153
Florida	14.6	35.3	11.0	16.5	0.000
Kansas	6.7	25.1	6.0	7.1	0.282
New York	35.2	47.8	38.2	33.6	0.027
Washington	14.3	35.1	17.6	12.6	0.001
Number of firms	2420		836	1584	

Table 2: Descriptives of outcome and covariates for STC and non-STC firms

Notes: ^a Compensated unemployment charges are total (STC and UI) benefits charged to firms are divided by full-time equivalent payroll (total wage plus 2 times the sum of the regular UI benefits and STC benefits, assuming a replacement rate of 50%). ^b Purely subsidized firms are those at the minimum/maximum UI tax rate. ^c Net balance is the difference between the average tax contribution of firms to the UI trust fund over 1991-1992 and the benefits paid out as a percentage of average taxable wage over 1991-1992.

	C	ommon Sul	pport	No	Common Sı	upport
	Weigł	nted Mean		Unweig	ghted Mean	
Covariates	\mathbf{STC}	Non-STC	ASDIFF	\mathbf{STC}	Non-STC	ASDIFF
Log of average wage/worker	10.151	10.164	0.031	10.154	10.134	0.043
Log of Employee per firm (in 1992)	3.500	3.508	0.006	3.525	3.118	0.332
UI Tax rate (normalized)	0.543	0.558	0.032	0.544	0.378	0.351
Net balance	-7.625	-8.191	0.050	-7.951	-3.551	0.382
Workforce stability indicator	0.466	0.464	0.004	0.461	0.335	0.258
Industry:						
Mining/construction/agric.	0.096	0.096	0.002	0.098	0.086	0.042
Manufacturing	0.501	0.495	0.013	0.502	0.424	0.158
Wholesale trade	0.072	0.076	0.013	0.071	0.094	0.086
Retail trade	0.066	0.063	0.009	0.065	0.082	0.067
Services	0.227	0.231	0.010	0.226	0.239	0.031
State:						
California	0.273	0.269	0.010	0.273	0.301	0.061
Florida	0.112	0.113	0.001	0.110	0.165	0.161
Kansas	0.061	0.059	0.010	0.060	0.071	0.047
New York	0.376	0.380	0.008	0.382	0.336	0.094
Average			0.014			0.151
<i>Notes:</i> Covariates are the main variables use teractions with the workforce stability indice ASDIFF is the absolute standardized different	d in computed in Weight	ting the probab ts, when applie s computed as i	ility of selection d to non-STC f n Morgan & Tc	into STC, i irms, are tho dd (2008).	along with quad e odds of selecti	ratics and in- on into STC.

Table 3: A check for balance in covariates between STC and non-STC firms

	(1)	(2)	(3)	(4)	(5)
β s.e.	0.165^{***} (0.058)	0.044 (0.060)	-0.082 (0.055)	-0.094^{*} (0.049)	-0.107^{**} (0.050)
Method	OLS + No Cov.	OLS + WSI	OLS + All Cov.	Matching	IPW
n	2420	2420	2420	2353	2353
<i>Notes</i> : In colu count for the u	mn 1, standard error (s.	e.) is robust. Stan	dard errors in all other	columns are boot.	strapped to ac-
	incertainty in estimating	group membership	based on workforce dyn	amics and the proh	bability weights

Table 4: Estimated constant STC effect on layoff rate

(sample attrition and selection). "OLS + No Cov" indicates no covariates are controlled for in a log-link Gaussian in a log-link Gaussian GLM. Matching is based on the local linear ridge matching technique, using a cross-validation bandwidth (in an Epanechnikov kernel) of 0.265. IPW is the inverse propensity weight estimator. Estimations of and "OLS + All Cov" indicates controlling for all covariates (used to estimate the probability of selection into STC) GLM, "OLS + WSI" indicates only the workforce stability indicator is controlled for in a log-link Gaussian GLM, the IPW and Matching are restricted to the region of common support. *p < 0.10, **p < 0.05, ***p < 0.01.

d/s0.150.200.050.10 -0.05 -0.084-0.100-0.091-0.086-0.072-0.10 -0.094-0.083-0.065-0.15-0.072-0.058-0.045-0.089-0.20 -0.082-0.061-0.042-0.022

Table 5: Sensitivity of the IPW estimate to an unobserved confounder

Notes: n=2353. Baseline $\hat{\beta} = -0.107$. The value *d* is the layoff effect in the program's absence of a simulated unobserved confounder, say 'skill'. The value *s* is the effect of the simulated unobserved confounder on selection into STC. Estimations are restricted to the common support region.

			·		
	Overall	Iı	ndustry	Pure	Subsidy
		Manuf.	Other	Yes	No
	(1)	(2)	(3)	(4)	(5)
		Cyclically stab	le and unstable f	firms	
Mean	-0.055	-0.046	-0.063	-0.186	-0.013
	(0.055)	(0.057)	(0.054)	(0.079)	(0.064)
P_{25}	-0.144^{**}	-0.120^{**}	-0.173^{**}	-0.268^{***}	-0.094
	(0.059)	(0.060)	(0.061)	(0.093)	(0.061)
P_{50}	-0.071	-0.061	-0.079	-0.218^{**}	-0.014
	(0.062)	0.064	(0.060)	(0.100)	(0.069)
P_{75}	0.047	0.056	0.043	-0.082	0.080
	(0.078)	(0.084)	(0.073)	(0.114)	(0.091)
		Cyclical	ly stable firms		
Mean	-0.003	0.000	-0.005	-0.075	0.009
	(0.075)	(0.075)	(0.076)	(0.121)	(0.082)
P_{25}	-0.082	-0.082	-0.083	-0.082	-0.082
	(0.077)	(0.078)	(0.076)	(0.121)	(0.077)
P_{50}	-0.024	-0.015	-0.030	-0.081	-0.005
	(0.077)	(0.079)	(0.076)	(0.123)	(0.086)
P_{75}	0.073	0.079	0.063	-0.079	0.088
	(0.100)	(0.103)	(0.099)	(0.127)	(0.112)
		Cyclically	y unstable firms		
Mean	-0.139^{**}	-0.137^{*}	-0.140^{*}	-0.248^{**}	-0.065
	(0.070)	(0.071)	(0.072)	(0.100)	(0.087)
P_{25}	-0.264^{***}	-0.258^{***}	-0.267^{***}	-0.280^{***}	-0.177^{**}
	(0.085)	(0.085)	(0.086)	(0.098)	(0.071)
P_{50}	-0.180^{**}	-0.177^{**}	-0.187^{**}	-0.259^{**}	-0.054
	(0.080)	(0.078)	(0.083)	(0.102)	(0.093)
P_{75}	-0.019	-0.024	-0.014	-0.227^{**}	0.064
	(0.096)	(0.099)	(0.096)	(0.110)	(0.118)

Table 6: Estimated heterogeneous STC effects on layoff rate

Notes: n = 2353 (906 are unstable firms and 1447 are stable firms). Standard errors are in parentheses and are bootstrapped to account for the probability weights (sample attrition and selection) and the uncertainty in estimating group membership based on workforce dynamics. Estimations are restricted to the common support region. Cross-validation bandwidths corresponding to the variables log average wage per worker, industry, workforce stability indicator, and tax rate (normalized) are 0.424, 0.500, 0.057 and 0.198, respectively. *p < 0.10, **p < 0.05, ***p < 0.01.



Figure 1: Compensated unemployment and employment trajectories over the period 1991-1992 for each latent grouping of firms identified from a finite mixture model. The lines represent the mean trajectory.



Figure 2: Density plots of the probability of selection into STC among STC and non-STC firms.



Figure 3: Density plot showing the distribution of the estimated heterogeneous STC effects on layoff rate.



Figure 4: Density plot showing the distribution of STC effects by industry.



Figure 5: Estimated STC effect on layoff rate as a function of log wage per employee among purely subsidized cyclically unstable firms. Black (gray) plots indicate estimated effects that are statistically significant (insignificant) at the 10% level.

References

- Abowd, J. M., Kramarz, F., & Margolis, D. N. (1999). High wage workers and high wage firms. *Econometrica*, 67(2), 251–333.
- Abraham, K. G. & Houseman, S. N. (1993). Job Security in America: Lessons from Germany. Washington, D.C.: The Brookings Institution.
- Abraham, K. G. & Houseman, S. N. (1994). Does employment protection inhibit labour market flexibility? Lessons from Germany, France, and Belgium. In Rebecca M. Blank (Eds): Social protection versus economic flexibility: Is there a trade-off? (pp. 59–93). Chicago, IL, University of Chicago Press.
- Abraham, K. G. & Houseman, S. N. (2014a). Proposal 12: Encouraging Work Sharing to Reduce Unemployment. In Melissa Kearney and Benjamin Harris, eds. *Policies to address* poverty in America, 129.
- Abraham, K. G. & Houseman, S. N. (2014b). Short-Time Compensation as a Tool to Mitigate Job Loss? Evidence on the US Experience During the Recent Recession. *Industrial relations: A Journal of Economy and Society*, 53(4), 543–567.
- Arico, F. R., & Stein, U. (2012). Was Short-Time Work a Miracle Cure During the Great Recession? The Case of Germany and Italy. *Comparative Economic Studies*, 54(2), 275– 297.
- Balducchi, D., Wandner, S., Goger, A., Miller, Z., Shetty, S., Agbayani, C., & Eucogco, J. (2015). Employer Views about the Short-Time Compensation Program: A Survey and Analysis in Four States, Impact International Report submitted to US Department of Labor Employment and Training Administration.
- Basu, A., Polsky, D., & Manning, W. G. (2008). Use of propensity scores in non-linear response models: the case for health care expenditures (No. w14086). National Bureau of Economic Research.
- Bellmann, L., Gerner, H. D., & Upward, R. (2012). The response of German establishments to the 2008-2009 economic crisis (No. 137). OECD Publishing.
- Boeri, T., & Bruecker, H. (2011). Short-time work benefits revisited: some lessons from the Great Recession. *Economic Policy*, 26(68), 697–765.
- Brookhart, M. A., Schneeweiss, S., Rothman, K. J., Glynn, R. J., Avorn, J., & Sturmer, T. (2006). Variable selection for propensity score models. *American Journal of Epidemiology*, 163(12), 1149–1156.
- Busso, M., DiNardo, J., & McCrary, J. (2014). New evidence on the finite sample properties of propensity score reweighting and matching estimators. *Review of Economics and Statistics*, 96(5), 885–897.
- Cahuc, P., & Carcillo, S. (2011). Is short-time work a good method to keep unemployment down? *Nordic Economic Policy Review*, 1(1), 133–165.

- Calavrezo O., Duhautois R. & Walkowiak E. (2009). The Short-Time Compensation Program in France: an Efficient Measure against Redundancies? Document de travail du CEE, no. 114.
- Calavrezo O., Duhautois R. & Walkowiak E. (2010). Short-time compensation and establishment exit: an empirical analysis with French data (No. 4989). IZA Discussion paper series.
- DeAngelo, H., & DeAngelo, L. (1991). Union negotiations and corporate policy: A study of labor concessions in the domestic steel industry during the 1980s. Journal of Financial Economics, 30(1), 3–43.
- Dehejia, R. H., & Wahba, S. (1999). Causal effects in experimental studies: Reevaluating the evaluation of training programs. *Journal of the American Statistical Association*, 94(448), 1053–1062.
- Dube, A., Freeman, E.,& Reich, M. (2010). *Employee Replacement Costs.* UC Berkeley: Institute for Research on Labor and Employment.
- Fraley, C., & Raftery, A. E. (2002). Model-based clustering, discriminant analysis, and density estimation. Journal of the American Statistical Association, 97(458), 611–631.
- Frolich, M. (2004). Finite-sample properties of propensity-score matching and weighting estimators. *Review of Economics and Statistics*, 86(1), 77–90.
- Gray, D. M. (1998). When Might a Distressed Firm Share Work? Evidence from the Short-Time Compensation Programme in France. British Journal of Industrial Relations, 36(1), 43–72.
- Greene, W. (2012). *Econometric Analysis* (7th Ed., International Ed.). Harlow: Pearson Education.
- Hall, P., Li, Q., & Racine, J. S. (2007). Nonparametric estimation of regression functions in the presence of irrelevant regressors. *The Review of Economics and Statistics*, 89(4), 784–789.
- Helper, S., Kreuger, T., & Wial, H. (2012). Why Does Manufacturing Matter? Which Manufacturing Matters? A Policy Framework. Washington, DC: Brookings Institution.
- Hijzen, A., & Martin, S. (2013). The role of short-time work schemes during the global financial crisis and early recovery: a cross-country analysis. *IZA Journal of Labor Policy*, 2(1), 1–31.
- Hijzen, A., & Venn, D. (2011). The role of short-time work schemes during the 2008-09 recession (No. 115). OECD Publishing.
- Ichino, A., Mealli, F., & Nannicini, T. (2008). From temporary help jobs to permanent employment: What can we learn from matching estimators and their sensitivity? *Journal* of Applied Econometrics, 23(3), 305–327.
- Imbens, G. W. (2004). Nonparametric estimation of average treatment effects under exogeneity: A review. Review of Economics and Statistics, 86(1), 4–29.

- John, K., Lang, L. H., & Netter, J. (1992). The voluntary restructuring of large firms in response to performance decline. *The Journal of Finance*, 47(3), 891–917.
- Kerachsky, S., Nicholson, W., Cavin, E. & Hershey A. (1986). An Evaluation of Short-Time Compensation Programs Occasional Paper 86-4. Washington, D.C.: U.S. Department of Labor and Mathematica Policy Research.
- Kruppe, T. & Scholz T. (2014). Labor hoarding in Germany: Employment effects of shorttime work during the crises. Institute for Employment Research (IAB) Discussion paper 17/2014.
- Li, Q. & Racine, J. S. (2007). Nonparametric Econometrics: Theory and Practice, Princeton University Press, Princeton.
- Li, Q. & Racine, J. S. (2010). Smooth Varying-Coefficient Estimation and Inference for Qualitative and Quantitative Data. *Econometric Theory*, 26(06), 1607–1637.
- Lilien, D. M., & Hall, R. E. (1986). Cyclical fluctuations in the labor market. Handbook of Labor Economics, 2, 1001–1038.
- MaCurdy, T., Pearce, J., & Kihlthau, R. (2004). An Alternative to Layoffs: Work Sharing Unemployment Insurance. *California Policy Review*, 1–11.
- Mattey, J., & Strongin, S. (1997). Factor utilization and margins for adjusting output: evidence from manufacturing plants. *Economic Review-Federal Reserve Bank of San Francisco*, (2), 3.
- Morgan, S. L., & Todd, J. J. (2008). A diagnostic routine for the detection of consequential heterogeneity of causal effects. *Sociological Methodology*, 38(1), 231–281.
- Needels, K., Nicholson, W., Kerachsky, S., Walsh, S., London, R. & McCanne, D. (1997). Evaluation of the Short-Time Compensation Programs. Final report, U.S. Department of Labor, Mathematica Policy Research and Berkeley Planning Associates.
- Van Audenrode, M. A. (1994). Short-time compensation, job security, and employment contracts: evidence from selected OECD countries. *Journal of Political Economy*, 76–102.
- Vandervieren, E., & Hubert, M. (2004). An adjusted boxplot for skewed distributions. COMPSTAT 2004, proceedings in computational statistics. Springer, Heidelberg, 1933– 1940.
- Vroman, W. (2013). Short-Time Compensation and Job Preservation. Washington, DC: The Urban Institute.
- Wooldridge, J. M. (2002). Econometric Analysis of Cross Section and Panel Data. Cambridge, MA: MIT Press.